Author's response to reviews

Title: Analysis of surgical and oncological Outcome in internal and external Hemipelvectomy in 34 Patients above the Age of 65 Years at a mean Follow-up of 56 Months

Authors:

Wiebke K Guder (wiebke.guder@ukmuenster.de)
Jendrik Hardes (jendrik.hardes@ukmuenster.de)
Georg Gosheger (georg.gosheger@ukmuenster.de)
Marcel-Philipp Henrichs (marcel-philipp.henrichs@ukmuenster.de)
Markus Nottrott (markus.nottrott@ukmuenster.de)
Arne Streitbürger (streitb@ukmuenster.de)

Version: 4 Date: 6 August 2014

Author's response to reviews:

Dear Ladies and Gentlemen,

as per your requirements the following changes to the document have been made:

1. The Title Page has been formatted to meet your requirements (email addresses of all authors have been added).

2. The Abstract has been restructured to meet your requirements.

3. The Main Text has been slightly formatted to meet your requirements. I.e. a list of abbreviations has been included.

4. Table Headings have been moved from below to above the respective Table as mentioned in Instructions for Authors.

We hope that our study will be considered for peer review and publication now. Please don't hesitate to contact us if other changes need to be made.

Sincerely,

W. Guder